**Author's response to reviews**

**Title:** Alterations of prolyl endopeptidase activity in the plasma of children with autistic spectrum disorders

**Authors:**

Naghi Momeni (naghi.momeni@spray.se)  
Berit M. Nordstrom (Berit.Nordstrom@med.lu.se)  
Vibeke Horstmann (vikeke.horstmann@geron.lu.se)  
Hassan Avarsaji (hassan_avarsaji@yahoo.com)  
Bengt V. Sivberg (Bengt.Sivberg@med.lu.se)

**Version:** 10  **Date:** 11 May 2005

**Author's response to reviews:**

We have updated the authors' e-mail addresses, and declared that the authors have no competing interests. Also, we have cropped the two figures and submitted an updated version of the main text body as well as we have removed the abstract from additional files.

Sincerely,

Bengt Sivberg